Author's response to reviews

Title: Schistosomiasis manifesting as a colon polyp: a rare finding

Authors:

Mona OSMAN (osmanmona@live.com)
Georges AFTIMOS (drg.aftimos@gmail.com)
Iyad ISSA (iyadissa71@gmail.com)

Version: 2 Date: 25 February 2014

Author's response to reviews:

1. the study design included in the title
2. co-authors’ email addresses included in the Title page.
3. Abstract section restructured
4. keywords included
5. Changed the 'Case Report' section header to 'Case Presentation'.
6. Changed the description of the patient’s gender to ‘female’.
7. Figure title and legend section included
8. Consent statement was included
9. Authors’ Contributions section included
10. Competing Interests section included
11. List of abbreviations included
12. No acknowledgement